

# **"ECONOMICS OF CULTURE, INSTITUTIONS, AND CRIME"**

Hosted by Fondazione Eni Enrico Mattei (FEEM)

Supported by

FP6 Priority 7 "Citizens and governance in a knowledge-based society" Project: Sustainable Development in a Diverse World"(SUS.DIV) (Contract No. CIT3-CT-2005-513438)

University of Padua Research Project "Economic analysis of crime and social interactions" (grant CPDA071899)

Fondazione Eni Enrico Mattei (FEEM)

Organized by SUS.DIV, FEEM, University of Padua and CEPR

Milan; 20-22 January 2010

# Advocatus, et non latro? Testing the Supplier-Induced-Demand Hypothesis for Italian Courts of Justice

Paolo Buonanno and Matteo M. Galizzi

We are grateful to the following institutions for their financial and organizational support: SUS.DIV, FEEM, University of Padua and CEPR.

The views expressed in this paper are those of the author(s) and not those of the funding organization(s) or of CEPR, which takes no institutional policy positions.

# Advocatus, et non latro? Testing the Supplier-Induced-Demand Hypothesis for Italian Courts of Justice \*

Paolo Buonanno UNIVERSITY OF BERGAMO UI

Matteo M. Galizzi UNIVERSITY OF BRESCIA

December 10, 2009

#### Abstract

We explore the relationship between litigation rates and the number of lawyers, in a typical supplier-induced demand (SID) frame. Drawing on an original panel dataset for the 169 Italian courts of justice between 2000 and 2007, we first document that the number of lawyers is positively correlated with different measures of litigation rate. Then, using an instrumental variables strategy we find that a 10 percent increase of lawyers over population is associated with an increase between 1.6 to 6 percent in civil litigation rates. Thus, our empirical analysis supports the SID hypothesis for the Italian lawyers: following an increase in their relative number, lawyers may exploit their informational advantage to induce clients to access to courts even when litigation is unnecessary or ineffective.

Keywords: lawyers, litigiosity, causality. JEL codes: F22, J15, K42, R10.

<sup>\*</sup>Contact information: paolo.buonanno@unibg.it, galizzi@eco.unibs.it. We gratefully thank Cassa Nazionale Forense for data on the lawyers enrolled in their pension scheme; Consiglio Nazionale Forense for data on the lawyers registered within each local bar association; Cassa Nazionale di Previdenza dei Dottori Commercialisti for data on the business consultants enrolled in their pension scheme; Consiglio Superiore della Magistratura for data on the judges within each first instance court of justice. We also would like to thank Silvia Dini, and her colleagues Gennuso and Brocco, of Consiglio Superiore della Magistratura for their valuable help in searching and getting the data. We thank Nicola Scalvini for research assistance. We are grateful Massimiliano Bratti, Francesco Fasani, Tommaso Frattini, Raffaele Miniaci, Samantha Vignati and to participants at seminars in Brescia, Alghero and Milan (MILLS) for many useful comments and suggestions. All errors are our responsibility. Paolo Buonanno gratefully acknowledges financial support from the University of Padua (research grant CPDA071899).

Sanctus Ivo erat Brito / Advocatus et non latro / Res miranda populo (Saint Ivo was Breton / A lawyer and not a thief / Something hard to believe). From the inscription in Treguier on the grave of Saint Ivo of Kremartin, patron saint of lawyers, a lawyer himself.

# 1 Motivation

On 30<sup>th</sup> June 2009, the day after the New York Court of Justice released the verdict to condemn Bernie Madoff to 150 years of conviction for his financial fraud,<sup>1</sup> Corriere della Sera, the most read and prestigious newspaper in Italy, reported the above vignette by its satiric columnist.<sup>2</sup> The vignette, show in its two halves a person in front of a court. In one half, it says: "US: six months for a 150 years sentence"; in the other, "Italy: 150 years for a six months sentence".

The vignette is probably the best summary we could find for the present work. Italy is quite rarely in the top positions of international rankings. However, in few other fields it can exhibit such an impressive collection of negative records as in the international comparisons for the legal system and the administration of justice. Not only Italy is the European country with the highest absolute numbers of both incoming and pending litigious civil cases into courts; but is also one of the European countries with the longest times to dispose of cases in first instance courts: 507 days on average for litigious civil cases (CEPEJ, 2008). It is perhaps not too surprising that Italy is the European country with the highest number of lawyers in absolute terms: in 2006 they were 170,143, compared to 138,104 lawyers in Germany, and 47,765 in France (CEPEJ, 2008). For instance, the city of Naples alone counted almost the same lawyers than in all England (*Consiglio Nazionale Forense*, 2009; CEPEJ, 2008). Furthermore, Italy is also the European country with the second highest number of lawyers per 100,000 inhabitants: 290 (only second to the 342 of Greece), compared to 168 of Germany, 76 of France, 22 of United Kingdom (CEPEJ, 2008).

What is even more striking is the impressive rate at which the number of lawyers has been increased in the last decades: according to data officially released by the agency managing the lawyers' pension scheme, lawyers have more than tripled in Italy, jumping from 42,366 in 1990 to 143,976 professionals in 2008 (+239.84%; *Cassa Nazionale Forense*, 2009). The trend is common to many other professionals in Italy, including accountants and business consultants (14,474 in 1992, become 48,109 in 2007, +232.38%; *Cassa Nazionale di Previdenza dei Dottori Commercialisti*, 2009). However, it is striking that such a jump in the number of lawyers occurred in a period in which the national GDP increased only by the 23.88% in real terms (IMF, 2009).

It is also in the period 1990-2007 that the litigation rate, measured by the number of

<sup>&</sup>lt;sup>1</sup>An updated version of the infamous "Ponzi scheme".

 $<sup>^2 {\</sup>rm Emilio}$ Giannelli, a lawyer and former Director of Legal Affairs of the third biggest Italian bank, Monte Paschi di Siena.

pending proceedings have further increased: incoming proceedings increased from 524.3 per 100.000 inhabitants in 1990 to 658.8 in 2007 (+25.7%), while pending proceedings passed from 804.8 per 100,000 inhabitants in 1990 to 1644.7 in 2007 (+104,4%).

The two trends in the increase in the number of lawyers and in the litigation are very likely to be related. Figure 1 below shows the correlation between the number of lawyers active in each court of justice and the litigation rate in that province. As it can be seen the positive correlation is quite sharp.

Figure 1: Provincial correlation between lawyers and total litigation rate (average 2000-2007)



These recent trends have recently attracted some attention from the literature on law and economics. Marchesi (2003) suggests that the higher litigation rate could have been caused by the existence of long trials: in fact, long waiting times to obtain a sentence give incentives to one of the party to breach contracts in the attempt to postpone due payments. Marchesi (2003) also observes that at least three, out of five, actors involved in a trial may have incentives to delay the time to obtain a sentence, namely the party that is in the wrong and the lawyers of both parties. In Italy, in fact, lawyers are typically paid according to the time and the number of judicial acts they are devoting to a case.

Some other studies have rather focused of the organization on the "supply" of justice in Italy and have mostly pointed to the low productivity by judges as the main responsible for high waiting times. Coviello et al. (2009), for instance, analyze the organization of two sections specialized in labor disputes in the of courts of justice of Milan and Turin, and observe that judicial offices working "in series" - that is, opening a new file only when the previous ones have reached a sentence - appear to be significantly associated with shorter waiting times than offices working "in parallel". Here we explore an alternative route. We explicitly test the hypothesis that the high litigation rate might be caused by an "excessive" access to courts by citizens. In particular, referring to the health economics literature, we propose an application of the supplier-induceddemand (SID) hypothesis to the case of Italian judicial system.

Under that perspective, an increase in the number of lawyers, and therefore in the density ratio between lawyers and population, might have forced lawyers to "induce" their clients to access more often the courts even when a judicial solution would have not been necessary or effective. Such an opportunistic behavior by the lawyers is possible because of the asymmetric information and imperfect agency relation between lawyers and their clients, and might be favored by the joint effect of tougher competition, caused by the new entrants, and of uniform minimum fees for service, required by the actual institutional context and the organization of legal professions in Italy. The induced, excessive demand for judicial resolution might thus have inflated the litigation rate.

In order to test this hypothesis, we have collected data from different official sources and built an original dataset on the 169 Italian courts of justice, considered in the period 2000-2007. In our OLS estimates, to mitigate for omitted variables bias we control for the number of judges in the court, for major socioeconomic and demographic characteristics, such as the levels of GDP, employment rate, population density, urbanization rate, education and social capital in the province. We exploit the panel structure of our data by estimating OLS models that allow for both province- and time-fixed effects.

Even after controlling for other determinants of litigation and for fixed effects, the number of lawyers across provinces could be correlated with the error term for several reasons. For instance, local higher rates of civil litigation in one court could attract lawyers into that province, which would bias our OLS estimates upward. In order to address this potential endogeneity problem, we adopt a Two-Stage-Least-Squares (2SLS) approach. What is needed is an instrument that is correlated with lawyers in the province and uncorrelated with unobserved factors that affects litigation. We consider two instrumental variables. The first is based on the fixed coefficient approach (Freeman, 1980; Bartik, 1991; Blanchard and Katz, 1992; Bound and Holzer, 2000; Card, 2001; Gould et al., 2002; Cortes, 2008): by considering the national levels of lawyers and lagged values of the local distribution of lawyers across provinces, we introduce an exogenous source of variation to be used as instrument for lawyers. The second instrument, based on the methodology proposed by Card (1993), is the average proximity of the province to the three faculties that offer law courses enabling to become lawyer.

Our main result is that the number of lawyers operating in a court does indeed exert a positive and statistically significant effect on the litigation rate. A 10% increase in the number of lawyers over population is associated with a 1.6% to 6% increase in litigation rates. This effect is robust across several specifications and checks, both on the control variables and the instruments. Our results thus support the SID hypothesis for the Italian lawyers. In particular, our evidence supports the idea that, following an increase in their relative number, lawyers may have taken advantage of the informational asymmetry to induce clients to access to the courts even when litigation would have not been necessary or effective.

Our work is organized as follows. Section 2 discusses the main characteristics of the organization of justice and of the legal professions in Italy. Section 3 reviews the literature on SID in health economics and applies it to the Italian judicial system. Section 4 describes the data and the variables. Section 5 reports the empirical strategy and the estimation results. Section 6 concludes with an overall discussion of the results.

# 2 The Organization of Justice in Italy

### 2.1 Courts of justice and judges

The administration of justice in Italy is organized in three degrees of courts of justice, that can be sequentially accessed. As a first level, there are 169 first instance courts of justice (*Tribunali di primo grado*). In many cases they cover the same geographic areas where the provincial administrative governments have jurisdiction. However, in several cases, especially in Piedmont and South Italy, there is more than one court of justice for province: some cover the most populated cities, some other follow the old jurisdictions of the former independent states existing before Italian unification in 1848. In 2008, a total number of 4,503 professional judges work in the 169 first instance courts of justice (CSM, 2009), divided almost equally between sections of the courts specialized in civil and penal justice (*Sezioni civili* and *Sezioni penali*, respectively).<sup>3</sup>

It is interesting to notice the evolution of the number of professional judges in the Italian first instance courts of justice: in the ten-years period 1998-2008, their number increased by 31,74%, passing from 3,418 to 4,503 (CSM, 2009). However, such a trend has not been monotonic. At the contrary, most the increase in the number of judges took place in 2000, 2002 and 2004, when more than 5,000 magistrates were active (CSM, 2009). That number, however, declined in the immediately following years.<sup>4</sup> Notice that, between 2000 and 2007, the period considered for our estimation, the number of professional judges in the Italian first instance courts of justice marginally declined (-3,49%) from 4,689 to 4,525.

 $<sup>^{3}</sup>$ In the larger courts of justice, typically the ones operating in the main cities where the second instance courts of justice are, some of the civil justice sections focus on the legal matters related to labor contracts (*Giudici del lavoro*). Administrative and tributary matters are instead dealt by specialized first instance courts of justice.

<sup>&</sup>lt;sup>4</sup>The reason behind this periodic increase, and then contraction, of the number of judges, is mainly due to the selection process to appoint the judges. The selection of judges is very strict and competitive: it requires a three-stage examination of the candidates at a national level that needs approximately two years to be concluded, and usually takes place every second year, following an official call for applications by the Ministry of Justice. Moreover, it can be noticed that, at least since 2004 on, where magistrates were 5,040, the total number of judges has been constantly declining (CSM, 2009). This is also reflecting the fact that, after 2004, no further selection to appoint new magistrates has taken place.

It is interesting to notice the evolution of the number of professional judges in the Italian first instance courts of justice: in the ten-years period 1998-2008, their number increased by 31.74%, passing from 3,418 to 4,503 (CSM, 2009). However, such a trend has not been monotonic. At the contrary, most the increase in the number of judges took place in 2000, 2002 and 2004, when more than 5,000 magistrates were active (CSM, 2009). That number, however, declined in the immediately following years.

The reason behind this periodic increase, and then contraction, of the number of judges, is mainly due to the selection process to appoint the judges. The selection of judges is very strict and competitive: it requires a three-stage examination of the candidates at a national level that needs approximately two years to be concluded, and usually takes place every second year, following an official call for applications by the Ministry of Justice. Moreover, it can be noticed that, at least since 2004 on, where magistrates were 5,040, the total number of judges has been constantly declining (CSM, 2009). This is also reflecting the fact that, after 2004, no further selection to appoint new magistrates has taken place.

The parties can appeal the sentence by a first instance court of justice and then access the second instance courts of justice. There are 29 second instance, or appeal, courts of justice (*Corti d'Appello*), usually one each Italian region, although the most populated regions have two. The regional areas over which a second instance court of justice has jurisdiction are called "districts" of appeal courts of justice (*Distretti di Corte d'Appello*). Finally, but only for procedural or formal matters, the parties can also appeal the sentence by the second instance courts of justice and then access the final degree of justice (*Corte di Cassazione*), which is only in Rome. It is important to notice that the distribution of professional judges across the three levels of justice is greatly skewed in favor of the first instance courts of justice, as many as 4,633 worked in the first instance courts (CEPEJ, 2008; CSM, 2009).<sup>5</sup> It is also important to bear in mind that, according to comparative statistics, Italy is among the European countries with the lowest number of professional judges sitting in courts per 100,000 inhabitants (11 in 2006, compared to an European average of 19.8).

To conclude the description of the organization of Italian courts of justice, we should mention the existence of non-professional judges, and, in particular, the institution in 1995 for civil, and 2002 for penal matters, of the figure of honorary judges of peace (*Giudici di Pace*): they are well respected and experienced law professionals that are temporarily appointed as honorary judges for a (renewable) period of four years, and serve as preliminary first instance courts for small claims such as controversies about houses, flats and gardens; goods and services of a value up to an amount of 2,582.28 euro; vehicles for a value up to 15,493.71 euro, among others.<sup>6</sup> There are currently 3,403 honorary judges of peace operating

<sup>&</sup>lt;sup>5</sup>Notice that this figure does not include the 2,231 (in 2006: CEPEJ, 2008) professional public prosecutors (*Procuratori della Repubblica*) operating in the separate and independent sections (*Procure*) within the first instance courts of justice.

<sup>&</sup>lt;sup>6</sup>The access to these honorary judges is much quicker, easier and cheaper for the parties than the one to

in Italy, distributed roughly in proportion of the professional judges in each first instance courts of justice (CEPEJ, 2008; Italian Ministry of Justice, 2009). The introduction of the honorary judges, aiming at turn away the simpler cases and smaller claims from the ordinary courts of justice already operating above capacity, it has been proved to be successful in reducing the number of files still pending in front of the courts of justice and in contributing to reduce the average time to reach a sentence.<sup>7</sup>

#### 2.2 Cases

To test the relationship between number of lawyers and litigation rate, we will focus on the civil cases processed by the first instance courts of justice, including the ones processed by honorary judges of peace. The choice to focus on litigation in civil courts only is mainly motivated by the crucial difference between civil and penal cases in the Italian justice. In fact, while a civil case may be started by any citizen, assisted by a lawyer, and is therefore a natural candidate to test the SID hypothesis, a penal file have to be started by a public prosecutor, thus leaving no room for inducement by lawyers.

The natural starting point is a comparative perspective within the European countries. Official statistics by CEPEJ (2008) gives Italy a number of international records. Considering 2006 statistics, for instance, Italy is the European country with the highest absolute number of incoming litigious civil cases into first instance courts: 2,825,453, compared to 1,104,828 cases of Germany and 1,688,367 of France (CEPEJ, 2008). Such a massive inflow of incoming cases, rather than an exception, is in line with the trend in the last fifteen-years period, and clearly contributes to keep the total number of pending civil litigious cases (at the end of 2006) at 3,687,965, again the highest absolute figure in Europe (CEPEJ, 2008). Even when such statistics are read in relative terms, the record of Italy remains greatly unchallenged: concerning 2006 data, Italy is the European country with the second highest number (after the Netherlands) of first instance incoming litigious civil cases per 100,000 inhabitants (4,809, compared to 2,672 of France and 1,342 of Germany: CEPEJ, 2008).

The high and constant number of incoming civil cases and the massive stock of pending cases have clearly an effect on the length of time necessary for a case to be disposed of by the courts. Indeed, the statistics on the average number of days that cases are outstanding or remain unresolved (*disposition time*) gives Italy some of the longest times to dispose of cases in first instance courts: 107 days were needed in 2006 for a non-litigious civil case, while as many as 507 days were needed for litigious civil cases, compared to 36 and 262 days, respectively, in France (CEPEJ, 2008).<sup>8</sup>

the ordinary first instance courts of justice; usually controversies are reconciled by proposing a compromise solution for the parties; sentences are rarely appealed to the ordinary courts of justice.

<sup>&</sup>lt;sup>7</sup>Nevertheless official comparative statistics show that Italy is still among the European countries with the lowest numbers of both non-professional judge per professional judge (1.1 in 2006, the fourth lowest figure) and non-professional judges per 100.000 inhabitants (12 in 2006, the third lowest figure: CEPEJ, 2008).

<sup>&</sup>lt;sup>8</sup>Among European Union countries, only Cyprus and Slovenia do worse than Italy.

#### 2.3 Lawyers

The organization of lawyers in Italy closely follows the organization of the courts of justice. Lawyers have to be enrolled in an official compulsory register (*Albo professionale*) that is hold and supervised by a local professional bar association (*Ordine degli avvocati*), to which the national law gives extensive legal prerogatives. There are 169 local professional associations, one every first instance court of justice. Bar associations are formed by all lawyers enrolled in the official register, who elect a council and a chairman (*Consiglio e Presidente dell'Ordine*). The latter are legally in charge of the supervision on the official register and, more generally, on the professional conduct of the associates. They also decide on all controversies among lawyers, and between lawyers and their clients, and, in order to enforce their supervision on the professional conduct of the associates, have some disciplinary powers, such as the suspension or expulsion from the official register.

The national law also regulates the criteria needed for lawyers being eligible to enroll into official registers. In fact, access to the legal profession in Italy requires a first degree in law (5 years), followed by a two-year apprenticeship in a lawyers' office (*Praticantato*). In order to obtain the official qualification as lawyers, successful candidates have then to pass a two-stage selection process, which is taking approximately one year.

Besides the local professional associations, a national council of lawyers (*Consiglio Nazionale Forense*) operates at the Ministry of Justice in Rome, to decide on controversies between local associations. The main prerogative of the national council of lawyers, however, is to set, every second year, all the payment tariffs and fees for service to be paid to lawyers, for civil, penal, administrative and tributary cases. Such decisions by the national council are subject to formal approval by the Ministry of Justice and are then legally binding for all lawyers and uniform across every local registers.<sup>9</sup>

In particular, the payment scheme set by the national council show some important features. First, unlike in other systems, contingent fees are not possible: the payment is always due and the client must pay the amount to the lawyer regardless of the outcome of the controversy. Secondly, the overall payment to the lawyer is directly proportional to the amount of legal services and acts provided by the lawyers. Moreover, even though it is possible to agree on a premium payment to the lawyer in case of a positive outcome, payments proportional to the value of the controversy are also explicitly ruled out. Finally, and importantly, the payment scheme as set by the national council is legally binding. In particular, no payment can be charged below the level set by the national council of lawyers, that thus serves as a compulsory minimum fee.<sup>10</sup> Lawyers, however, are free, at a large

<sup>&</sup>lt;sup>9</sup>In particular, the Ministry of Justice decree 127/2004 specifies that the payment to the lawyers (*parcella*) is constituted, on the top of the reimbursement of all expenses, by two parts: the tariffs (*onorari*), and the fees for services (*diritti*).

 $<sup>^{10}</sup>$ The existence of such minimum compulsory fees for legal services has been challenged by the decree 223/2006 (the so-called Bersani decree) aiming at their elimination. Although the national council of lawyers opposed the decree and tried to limit its applicability, the decree became finally operating from 2008. This also

extent, to ask any higher payments than the amounts set by the national council.

Data on the number of lawyers are available from two different sources. On the one hand, the national council of lawyers does not provide yearly data on the number of lawyers enrolled into each local register, while these statistics are available only every two-three years. The main reason why this source of data is scarcely reliable for an empirical investigation does not lie in their incomplete availability, though. In fact, by their nature, the registers from the local bar associations include all graduates in law who, at some point, got the professional qualification as lawyers, but actually no longer work, or even have never worked, as lawyers. This is because in Italy who got the professional qualification as a lawyer has the right to remain in the official register regardless of whether, and how effectively, is working as lawyer. The local registers thus include many professionals who are actually working only occasionally, or have soon abandoned the profession for different career patterns.<sup>11</sup>

As the two sources of data differ, there is a discrepancy between the number of lawyers registered to the official registers (*Albo Avvocati*) and the number of those enrolled into the professional pension scheme (*Cassa Forense*). In particular, the latter underestimates the official number of lawyers.<sup>12</sup> As empirical estimates are concerned, issues may arise only if the difference between the two measures would change over time within provinces. Data available show that the ratio between the two measures is approximately constant across time within provinces. Therefore, controlling for both province- and time-fixed effects allows to adequately deal with the potential measurement error.<sup>13</sup>

Data show that in less than 20 years, the number of lawyers enrolled in the national pension scheme more than tripled, jumping from 42,366 professionals in 1990 to 143,976 in 2008 (+239.84%). The total number of lawyers doubled during the 90s and, between 2000 and 2007 (the period considered for our estimation), experienced an equally impressive increase (+65.48%), from 82,637 to 136,750 professionals (Cassa Nazionale Forense, 2009).

As already mentioned, European statistics show that in 2006 Italy was the European country with the highest absolute number of lawyers, and the highest number of lawyers per professional judge: 26.4, compared to 7.1 of France, 6.9 of Germany and 3.2 of United Kingdom. Italy was also the European country with the second highest number of lawyers per 100,000 inhabitants: 290 (only second to the 342 of Greece), compared to 168 of Germany,

explains the 2000-2007 time period we have considered for our empirical estimation. The Italian Parliament is currently debating a new law reintroducing compulsory minimum fees for lawyers' services.

<sup>&</sup>lt;sup>11</sup>How serious is this issue is confirmed by the recent, fiercely debated, intention by the national council of lawyers to exclude from the local registers any professional who is not earning from the lawyers' activity a minimum level of income. The statistics published in the official publication by Cassa Forense (Biancofiore, 2009; Donella, 2009) show that 47.605 lawyers enrolled in the official registers declared in 2008 no income from the legal profession.

 $<sup>^{12}</sup>$ In 2008, 53,969 lawyers were enrolled into the official registers but not in the national pension scheme. The discrepancy between the two measures appears to be specially large in southern Italian provinces. Many possible explanations can be put forward. Since the enrollment into the pension scheme is compulsory only from a minimum threshold of earned income, it may be more difficult for lawyers in southern Italian regions to earn a sufficiently high income. Tax evasion may be an alternative explanation.

 $<sup>^{13}\</sup>mathrm{For}$  a detailed analysis of the discrepancy between these two sources see Donella (2009)

76 of France and 22 of United Kingdom (CEPEJ, 2008).

# 3 The Supplier-Induced-Demand Hypothesis

Direct inspection of official statistics shows that a greater supply of lawyers went hand-in-hand with an increased litigation rate in Italian courts of justice. For instance, in the 1990-2007 period, while the number of lawyers almost tripled, the number of files pending in front of the first instance courts of justice increased by 104.4%. Looking at provincial level (see Figure 1), it emerges a strong correlation between the number of lawyers and the litigation rate in the local courts, this generally holding for several definitions of litigation.

The mere fact that consumption of legal services, and therefore, litigation rate increases with the number of lawyers, however, it is consistent with several explanations. For instance, if the market for lawyers were competitive, then an increase in the number of lawyers should lead to an outward shift of the market supply function. Combined with a downward-sloped demand curve, this predicts an increase in the number of traded legal services, paired by a fall in their fees. The effect on total expenditure is ambiguous and depends on the price elasticity of demand. Under this perspective the higher litigation rate is simply the demand reaction to lower fees consequent to an increase in the supply, and is therefore nothing but a market adjustment.

The argument based upon this market explanation, however, cannot apply to the case of Italian lawyers. In fact, the Italian institutional context has greatly favored a substantial rigidity of lawyers' fees. The main cause for lawyers' fees failing to decline is related to the above discussed legal privilege by the national council of lawyers to set a minimum fee for legal service, which is binding and uniform across all Italian courts of justice.<sup>14</sup> The existence of such minimum fees may not only represent a lower bound for the attempts to decrease prices, but also serve as a salient reference price for lawyers when setting their fees,<sup>15</sup> thus curbing price competition, as also pointed out by the Italian antitrust authority in several occasions (Autorità Garante della Concorrenza e del Mercato, 1997, 2009).

As we will argue below, the joint presence of a compulsory minimum fee, set by the national council of lawyers, and of a sufficiently large share of new entrants in the market who are willing to provide legal services at the minimum fee, are conditions that makes the above market explanation hard to believe for the case of Italian lawyers.

An explanation alternative to physiological market adjustment can in fact be found in the peculiarities of the lawyer-client relationship. One of the key ingredient of this relationship is

<sup>&</sup>lt;sup>14</sup>A similar relation has been observed in health economics literature between increased supply of physicians and increase in utilization of health care services, in spite of non declining prices for medical services: Fuchs (1978) observe it in US even in presence of higher fees; Adam (1983), Breyer (1984) and Breyer et al. (1986) detect it in some German Lander where fees were fixed at a uniform rate across regions; Grytten et al. (1990) observe it for the demand for dental services in Norway, where a national fixed price was in place.

<sup>&</sup>lt;sup>15</sup>For a model and some experimental evidence of the role of reference prices in coordinating pricing strategies and favoring collusive behavior see Miraldo (2009) and Galizzi and Miraldo (2009).

the clients' incomplete information about their true needs. Typically clients who consults a lawyer only knows that they require some legal assistance, because, for instance, they are not satisfied with some contractual obligation or extra-contractual relation. Typically, the client tends to trust the better informed lawyer, to delegate to the latter the choice of the exact legal and juridical instruments, and, more often than not, to follow the lawyer's advices and recommendations. Services from the legal professions, such as the ones by doctors, financial experts or art consultants, can in fact be viewed as credence goods, in the sense of Darby and Karni (1973): even after having used such services, due to the informational asymmetry clients cannot verify whether professionals acted in their interests.

This remains without consequences as long as lawyers act as perfect agents for the clients, choosing what the clients would have chosen if they had possessed all necessary legal knowledge. It does become a problem, however, as long as the lawyers' decisions on behalf of the clients are influenced by their own interests. Lawyers, in fact, are not merely agents but also providers and sellers of legal services. Italian lawyers are not an exception. Indeed, interests of lawyers and clients are potentially conflicting in two aspects.

First, following an increase in the number of lawyers, such as the one observed for Italy in the 1992-2007 period, lawyers may have faced the threat of a strong competitive pressure in terms of fewer clients in their portfolio. Lawyers, however, may envisage the goal to secure their own full employment or to maintain their previous, higher, income level. This would be consistent with the target income hypothesis by Evans (1974), originally applied to physicians, by which professionals have a desired level of income that they strive to achieve, or to restore, whenever actual income falls below the target. In such a case, following an enhanced competitive pressure and a contraction in clients' portfolio, lawyers may have been tempted by manipulating the information provided to their clients in order to induce them to unnecessarily access to courts. This potential source of conflict of interests is favored by the Italian legislation by which clients cannot pay to lawyers fees contingent on the outcome of the litigation.

Secondly, once started a case, lawyers have interest in inducing the client to request a large number of acts and legal services. In fact, potential conflict of interests are exacerbated by the fact that in Italy the payment scheme designed by the national council of lawyers is not only proportional to the time and effort spent in a case, but also integrated by a plethora of fees and tariffs for each legal act or service supplied.<sup>16</sup>

Therefore, these potential sources of conflict of interest make the agency relationship between lawyers and their clients an imperfect one, and provide to lawyers incentives to exploit their informational advantage in their own interests, by inducing not necessary, or inefficient, access to courts or excessive provision of legal services. Due to asymmetric information, the

 $<sup>^{16}</sup>$ Van De Voorde et al. (2001) found some indirect evidence of suppliers-induced demand in the context of the Belgian national health system, characterized by excessive supply of doctors and by a fee-for-service system.

demand by clients can no longer be treated as sovereign, and is, at the contrary, induced by suppliers.

The supplier-induced-demand hypothesis in the specific case of Italian lawyers can also be illustrated by mean of a simple graphical representation (Figure 2). Imagine that clients' demand for legal services (L) negatively depends on the level of lawyers' charged fees (F). For the sake of simplicity, suppose the initial demand function is linear in fees and can be represented as  $D_0$ .

Figure 2: supply and demand functions with minimum fee



Denote  $\underline{F}$  the minimum fee for service set by the national council of lawyers. As discussed above, this level is uniform across all districts of justice and legally binding for all Italian lawyers, in the sense that no lawyer in Italy can charge less than F for providing legal services, while is free, at some extent, to charge fees higher than F. The supply function of legal services by Italian lawyers can therefore be expressed as a schedule of the following type:

- for any  $F < \underline{\mathbf{F}}, L = 0;$
- for  $F = \underline{F}$ , L is any value within  $[0, \underline{L}_0]$ ;
- for any  $F > \underline{\mathbf{F}}, L = \underline{\mathbf{L}}_0 + \alpha F$ ;

with  $\alpha > 0$ , and where  $L_0$  stands for the proportion of lawyers in the market that accept to supply legal services for a fee equal to the minimum fee set by the national council.

The number  $\underline{L}_0$  of lawyers working for the minimum fee can be thought as a function of some underlying characteristics of the organization of the legal professions in Italy. For instance, one can think to  $\underline{L}_0$  as the share of young lawyers in the market. In fact, professionals that have just gone through all the long selection process, and have recently qualified to work as lawyers, can typically act as aggressive entrants in the market and, in order to gain experience and build a clients' portfolio, may therefore accept to work for fees equal to the minimum set by the national council.<sup>17</sup> Alternatively, one can think to  $\underline{L}_0$  as reflecting the existent competitive pressure in the provision of legal services from suppliers other than lawyers. In Italy for instance, workers and consumers can access trade unions and consumers' associations, respectively, in order to get legal assistance for small claims. On the other hand, while large companies typically have internal legal offices dealing with most standard issues, small and medium enterprises are also able to access other categories of consultants (such as tax advisors, experts in labor and pensions issues, among others), or even consulting services organized and provided by local chambers of commerce and branches of the business associations they are member of.<sup>18</sup>

The supply schedule for legal services can thus be represented as a kinked, piecewise linear, upward-sloping function with the shape represented by  $S_0$  in the figure 2. In fact, for any fee lower than <u>F</u>, no legal service is provided. For fees exactly equal to the level of the minimum fee, there is a proportion <u>L</u><sub>0</sub> of lawyers that accept to supply legal services. The  $S_0$  function represented in figure 2, then assumes the remaining share of lawyers offer an amount of legal services which is directly (and linearly) increasing with the charged fees, as in a standard upward sloping supply function.

As usual, the intersection between demand and supply function provides the equilibrium level of legal services and fees in the market. In our case these crucially depend on the initial relative position of  $D_0$  and  $\underline{\mathbf{L}}_0$ . In fact, when the share  $\underline{\mathbf{L}}_0$  of young lawyers is relatively low compared to the demand schedule, intersection typically occurs in correspondence of the upward sloped piece of  $S_0$ , and the equilibrium market fee is higher than the minimum legal fee,  $F_0^* > \underline{\mathbf{F}}$ , as in figure 2. On the other hand, when there are relatively many lawyers in the market that accept to work for the minimum fees, equilibrium fees in the market coincide with the minimum level set by the national council,  $F_0^* = \underline{\mathbf{F}}$ , as in figure 3

Clearly, the effects of an increase in the number of active lawyers in the Italian market, and the likelihood of eventually support the SID hypothesis, ultimately depends on the initial market equilibrium. In fact, imagine that, following, for instance, the entry of a flow of young professionals, an increase occurs in the level of active lawyers in the market. This can be

<sup>&</sup>lt;sup>17</sup>Despite the lack of official data, anyone can find support to this conjecture from an ecdotal experience and direct conversations with young lawyers.

<sup>&</sup>lt;sup>18</sup>Such as *Confindustria*, the Italian equivalent of the Confederation of Business Industry in United Kingdom.

Figure 3: Equilibrium market price equal to the minimum fee



represented by an outward shift of the supply function from  $S_0$  to  $S_1$ : in particular, as more young lawyers enter the market, the number of lawyers that accept to work for the minimum fees also shifts outwards, from  $\underline{L}_0$  to  $\underline{L}_1$ .

The SID hypothesis assumes that, following the entry of new professionals and the shift in the supply function, lawyers in the market would be tempted to exploit their asymmetric advantage in order to induce their clients to demand unnecessary or ineffective legal services. The increase in supply would thus induce also an increase in demand, possibly by a lower extent. Graphically, the demand schedule would also shift outwards, from  $D_0$  to  $D_1$ . These shifts would imply that, in the new market equilibrium, clients buy a larger quantity of legal services from lawyers. Therefore, under the SID hypothesis, an increase in the number of lawyers in the market would be typically associated to a larger access to courts of justice, and to a higher litigation rate.

However, to find a full support to the SID hypothesis, the increase in the litigation rate should be exclusively due to the artificially inflated demand induced by lawyers. In particular, to fully accept the SID hypothesis, one should also rule out the possibility that the higher litigation rate derives from a demand that physiologically increases because of a drop in the fees. Therefore, the SID hypothesis is compatible only within some specific initial market equilibrium. In particular, imagine that the initial share of lawyers working for the minimum fees  $\underline{L}_0$  is relatively low, and that the initial intersection with demand occurs at a point in the upward-sloped piece of the supply function. As in figure 4, in the initial market equilibrium an amount  $L_0^*$  of legal services are bought at an equilibrium fee  $F_0^*$ , higher than the minimum fees. In such a case, the induced demand by lawyers following the entry of new professionals may not be the only explanation to an observed higher litigation rate. In fact, following a shift from  $S_0$  to  $S_1$  due to a higher number of lawyers active in the market, and a partially induced outwards shift in demand, the new market equilibrium implies a higher number of traded legal services  $L_1^* > L_0^*$ , but also a lower equilibrium fee  $F_1^* < F_0^*$ . Therefore, even in presence of some inflated demand induced by lawyers, it cannot unambiguously ruled out the possibility that, at some extent, clients have indeed demanded more legal services because fees have been reduced by increased competition. This, in fact, would also be compatible with standard market adjustment.





On the other hand, imagine there are relatively many lawyers in the market that accept to work for the minimum fees, so that intersection with demand occurs at a point in horizontal piece of the supply function, and equilibrium fees in the market coincide with the minimum level set by the national council. In such a case the SID hypothesis may be fully supported as an explanation for an observed higher rate of access to courts. In fact, following the entry of more lawyers in the market, lawyers can, through their advises to clients, artificially inflate the demand for their services to match the higher supply, shifting demand outwards from D0 and D1. This leads to a new market equilibrium, such as the one illustrated in figure 5, in which a larger amount of legal services is bought,  $L_1^* > L_0^*$ , even in presence of unaffected fees for services, that remain fixed at the level of the minimum fees F.

Figure 5: Increase in supply and induced demand



Therefore, the SID hypothesis can be supported as a theoretical explanation for an observed increase in the litigation rate after an inflow of new lawyers into the market. In particular, the SID hypothesis turns out to be a particularly convincing explanation when there is a relatively high share of professionals who accept to provide legal services at the minimum fees set by the national council of lawyers. This is very likely to be the case in Italy, where, as discussed above, an extraordinarily high number of young professionals have recently entered the market.

One can still argue that SID is not the only possible explanation accounting for the positive correlation between the number of lawyers and the litigation rate, even in presence of not declining fees. In particular, a typical argument that can be put forward is one based upon a reverse direction of causality. When choosing a location for their office, in fact, young lawyers may try to assess the areas where demand will be high enough to guarantee a sufficient level of expected revenue. Areas and cities with high demand for the legal services should thus attract more lawyers than those with low demand. Consequently, a high per capita utilization of legal services leads to a high lawyers density on average, and not vice-versa, as assumed by the SID hypothesis. This argument puts emphasis on a potentially serious endogeneity problem, between the number of lawyers and the demand for legal services. However, applied econometrics can deal with the endogeneity problem, for instance through the instrumental variables two-stages technique that we employ in our estimations. Therefore, the definite answer to such a reverse causality question is simply an empirical matter, that we address more in detail in the next Sections.

For what concerns the literature on law and economics, to the best of our knowledge, no study has yet directly and explicitly tested the SID hypothesis for legal services. However, some indirect evidence in support of the SID hypothesis for lawyers has been found by Ginsburg and Hoetker (2006) for the case of Japan, and by Sobbrio et al. (2009) in the case of Italy.

On the other hand, among health economists interested in physicians' incentives, a number of studies have already attempted to empirically test the SID hypothesis. In general, several studies have provided some evidence in support to the hypothesis. For instance, Fuchs (1978), using cross-section data on US surgeons, found that an increase in 10% surgeon density led, ceteris paribus, to a 3% increase in the frequency of surgery. Tussing (1983) used data from Irish general practitioners and found that the number of visits and the likelihood that a visit was initiated by the physician were significantly positively correlated with the number of physicians per population. Adam (1983), Breyer (1984) and Breyer et al. (1986) analyzed regional German data between 1977 and 1982 and, using multi-equation models to simultaneously explain physician density and per capita expenditure in medical services, found statistically significant elasticity of per capita expenditure to physician density (estimated 0.1-0.4 elasticity). Kraft et al. (1986) using data from a medical practice in the Canton of Berne, Switzerland, found a statistically significant, positive, correlation between expenditure per medical case and physician density, and, after a Hausman test, concluded that the latter was to be considered exogenous. Grytten et al. (1990) supported the SID hypothesis for the Norwegian market for dental services, where, in a context of fixed, uniform fees, both the demand and the expenditure for dental services increased as the population/dentist ratio decreased. Grytten et al. (1995), using Norwegian data, found some evidence of SID for diagnostic laboratory tests requested by doctors. Gruber and Owings (1996), analyzed data from the US National Hospital Discharge Survey in the 1970-1982 period (on 200.000 discharges from 400 hospitals) and found evidence in favor of the SID hypothesis in the number of Caesarian deliveries, more lucrative, and requiring a lower workload, to US obstetricians than natural child birth deliveries. Van De Voorde et al. (2001), by estimating the effect of an increase in co-payment rates on patients' out-of-pocket price elasticity, indirectly found some evidence of inducement by general practitioners in Belgium, characterized by an excessive supply of doctors and by a physician remuneration scheme based on fees for services.<sup>19</sup>

<sup>&</sup>lt;sup>19</sup>The validity of the empirical results by Van De Voorde et al. (2001) has been questioned by Cockx

The findings of the health economics literature are not unanimous, though. A number of papers that tested the SID hypothesis, did not find strong evidence in support of it. Rossiter and Wilensky (1984), for instance, found little support for the SID hypothesis from the data on expenditure for health services by a representative sample of US population, contained in the National Medical Data Expenditure Survey. Grytten et al. (1995) used survey data on physician-patient contacts for a representative sample of the Norwegian population and did not find any sign of SID on the number of physician-initiated visits, although they found some evidence of it in the number of diagnostic laboratory tests required by physicians. Madden et al. (2005) studied the effect of a change in the reimbursement system for Irish general practitioners on the utilization of their services and found ambiguous evidence on SID.

The current debate in the health economics literature is mostly related to the choice of the empirical strategy. Indeed, an accurate econometric analysis testing the SID hypothesis should convincingly deal with the potential endogeneity and causality issues, and, ultimately, on the selection of good instruments to be used in the instrumental variable estimation. We will discuss our empirical methodology in the next Section.

# 4 Data Description and Panel analysis

Our balanced panel dataset comprises annual observations for the 103 provinces over the period from 2000 to  $2007.^{20}$  Data regarding civil proceedings and lawyers are collected at the level of courts of justice and then aggregated at the level of the 103 Italian provinces.<sup>21</sup>

Civil litigation data come from the civil justice statistics recorded by the courts of justice. These statistics are published yearly by the Italian Statistical Institute (ISTAT) and allow a disaggregation by province and type of litigation. In particular, we distinguish among different forms of litigation: first-instance ordinary civil proceedings in front of civil courts of justice (courts of justice); first-instance ordinary civil proceedings in front of honorary judges of peace (judges of peace); total litigation rate, given by the sum of the previous two (total litigation rate); and litigation for compensation (compensation).

The main explanatory variable is the number of lawyers over population at provincial level (lawyers). As discussed above, we use the total number of lawyers enrolled in the pension scheme, divided by local professional register, as obtained by *Cassa Forense*, the agency managing the lawyers' pension scheme. In order to control for the lawyers-judges ratio and for a proxy of productivity in each court, we add as explanatory variable the numbers of judges over population at provincial level in each year (judges).

and Brasseur (2003), who argued that the authors' estimated price elasticity ignored the substitution effects induced by the change in relative prices of physician services.

 $<sup>^{20}</sup>$ The choice of the period considered for our empirical estimation is motivated by the fact that, from 2008 on, the minimum fees set by the national council of lawyers are no longer considered as compulsory, in light of the above discussed decree 223/2006. Moreover, 2008 data are not available.

<sup>&</sup>lt;sup>21</sup>Italian provinces correspond to the NUTS3 areas and are comparable in size to US counties.

Our dataset also includes a set of socioeconomic and demographic variables that are likely to be correlated with litigation rates. Demographic variables include the population density in the province (*density*), and the concentration index (concentration), calculated as the ratio between the population residing in the provincial administrative city over the population in the rest of the provincial area. We include these variables as in more densely populated areas the number of social interactions and, thus, the potential conflicts and dispute are greater. Turning to the socioeconomic variables, we include the (log of) real GDP per capita, (*GDP per capita*), and the employment rate, (*employment*), which measure the size and development of provincial economy, both of which have been proved to be potentially correlated to the litigation rate (Posner, 1997; Hanssen, 1999; Clemenz and Gugler, 2000; Ginsburg and Hoetker, 2006).

It may be argued that also the level of education of the population may affect the level of civil litigation, although the direction of such an effect seems more ambiguous. In fact, on the one hand, better educated population may also be more aware of their rights and, possibly, more inclined to pursue their rights by accessing the courts. On the other hand, better educated population may also be more aware of alternative ways of dispute resolution, or more informed about the high costs and long time horizons necessary to conclude a dispute in courts, and therefore, more discouraged to access courts of justice. In order to check the effect and, if any, the direction of education on litigation, we include the percentage of population with high school diploma as a proxy for education (*high school*).

Finally, the level of civil litigation may be affected by the level of social capital in the province. For instance a higher level of social capital may reduce civil litigation since individuals are more respectful of contracts and laws and also because may favour an informal resolution of civil disputes. Following Buonanno et al. (2009) we include, as a measure of social capital, the density of associations (*associations*), namely the number of recreational, cultural, artistic and sportive no-profit associations, every 100,000 inhabitants at provincial level in each year.

Our list of control variables is likely to be incomplete, since it is impossible to control for all factors affecting civil litigation. Thus, to control for unobserved factors, we exploit the panel structure of our dataset and we include both province- and year-fixed effects.

Detailed definitions and sources are presented for all of the variables in the Appendix. Descriptive statistics are presented in Table 1, while Table 2 reports the correlation matrix among all dependent and explanatory variables. As already mentioned, the correlation between lawyers and civil litigation is positive for all types of civil litigation considered.

Our main estimating equation is

$$litigation_{it} = \beta lawyers_{it} + \gamma' X_{it} + \phi_i + \phi_t + \varepsilon_{it} \tag{1}$$

where  $litigation_{it}$  is the log of the civil litigation rates recorded by the civil courts in province

*i* during year *t*;  $lawyer_{it}$  is the number of lawyers over population;  $X_{it}$  is a set of control variables;  $\phi_i$  and  $\phi_t$  are province- and year-fixed effects; finally,  $\varepsilon_{it}$  is an error term.

The set of observables  $X_{it}$  comprises demographic and socioeconomic determinants of civil litigation discussed and presented above.

OLS estimates on equation (1) are presented in table 3 and suggest that the different measures of litigiosity are significantly correlated with the incidence of lawyers in the population. This relationship is overall robust across the different definitions of litigation rates and even across alternative specifications of determinants and controls of civil litigation. According to these findings, a 10% increase in the number of lawyers over population is associated with a 3% increase of total litigation rate, a 1.6% increase of litigation rate in civil courts of justice, a 4% increase of litigation rate in front of honorary judges of peace and 5% increase in litigation related to compensation.

Turning to the control variables, it emerges that GDP, employment rate and the number of judges in the court do not exert a significant effect on local litigation rates. Our measure for education is negatively correlated to litigation rates, even if not always significant. Population density is negatively and significantly correlated with our dependent variables, while concentration index exerts a positive and significant effect on litigation rates. Finally the level of social capital does not present consistent patterns and is sensitive to the type of litigation rate considered.

However, there could be several reasons why the number of lawyers over population is systematically correlated with litigiosity, some of which may not be adequately captured by control variables. Therefore, identifying causality requires a source of exogenous variation in the number of lawyers, an issue that we tackle in the next section.

# 5 Causality

Even after controlling for other determinants of litigation and for time and province fixed effects, the number of lawyers across provinces could be correlated with the error term for several reasons. As discussed in Section 3, the SID hypothesis implies that an increase in the number of lawyers and, therefore, of the degree of competition among legal professionals, could lead lawyers to exploit their informational advantage to induce clients to demand legal services and to access courts even when that would not be necessary or effective. In Section 3, however, we have already mentioned an alternative explanation that may potentially explain a statistical relation between the number of lawyers and the litigation rate. In fact, it is in theory possible that the local higher rate of civil litigation in a court had attracted lawyers to that area, in the attempt to serve a share of the high demand.

#### 5.1 Methodology

In order to address the endogeneity issue we need some variable that is a good measure of the number of lawyers in a court but is exogenous to changes in the local litigation rates. Here we consider two instrumental variables. The first is based on the "fixed coefficient" approach and uses national levels of lawyers and lagged values of the local distribution of lawyers across provinces. The second instrument, based on the methodology proposed by Card (1993), uses the average proximity of the province to the three closest law schools.

The idea (and name) for the fixed coefficient approach comes from Freeman (1980), where the author empirically assessed the ability of the fixed coefficient "manpower requirements" model to explain future changes in employment. Following Bartik (1991), Blanchard and Katz (1992), Bound and Holzer (2000), Card (2001), Gould et al. (2002) and Cortes (2008), our first instrument (*predicted lawyers*) builds on the interaction between two sources of variation, which are exogenous to changes in local markets for lawyers: the initial distribution of lawyers across provinces and the trend, at a national level, in the number of lawyers.

Formally, the instrument for the number of lawyers in province i and year t is defined as a province-specific weighted average of national lawyers:

$$\widehat{lawyer_{it}} = \bar{\omega}_{iT} * lawyer_t \tag{2}$$

where  $lawyer_{it}$  is the predicted number of lawyers in province *i* and period*t*,  $lawyer_t$  is the national number of lawyer in year *t* and  $\omega_{it}$  is the share of lawyers in province *i* over the total national number of lawyers in some base year *T*. In details, for each province *i*, we compute the provincial share of lawyer in some base year *T* as:

$$\omega_{iT} = \frac{lawyer_{iT}}{\sum_{i} lawyer_{iT}} = \frac{lawyer_{iT}}{lawyer_{T}} \tag{3}$$

In other words, the provincial share of lawyers in province i and some base year T indicates the percentage of lawyers residing in province i over the national number of lawyer in some base year T.

One concern is that the initial distribution of lawyers across provinces may reflect expected changes in the local litigation rate. To lessen any fear that the initial provincial share of lawyers is itself endogenous, we use the 1992 provincial distribution instead of the 2000 one. Moreover, as robustness checks, we use as initial provincial shares of lawyers, the distribution across provinces in subsequent years: 1993, 1994 and 1995.

The above instrument is exogenous with respect to local litigation rates if both its components are exogenous. As already mentioned, the provincial shares of lawyers in the base year  $T(\bar{\omega}_{iT})$  is exogenous because measured in a period T preceding the time span to which our data on litigation refer. On the other hand, the national trend in the number of lawyers are mainly driven by demand for lawyers at a national level, and is therefore sufficiently exogenous to provincial changes in the local markets for lawyers. This predicted measure conveys the number of lawyers in each province that would have been observed if the distribution of lawyers across provinces had not changed since period T.

We also consider a second instrument (*proximity*) capturing the presence of law schools accessible from a province, as this could clearly be related to the individual choice of subject at university. Following Card (1993), we use as an instrument for the number of lawyers the provincial proximity to schools where to pursue a degree in law. In particular, the instrument we use is the average proximity between the three closest universities offering a degree allowing to obtain a qualification as a lawyer. The latter aspect is particularly relevant in Italy, since not only the Faculties of Law but also the ones of Economics and Political Sciences may offer degrees in law that allow student to go through the selection process to become a lawyer. As a robustness check, we also consider the average proximity between the thee closest Faculties of Law to a province.

The Italian university system has two peculiar features in favour of our instrument: i) there is very low mobility of students across cities and universities;<sup>22</sup> ii) since Italian universities, except rare exceptions, are public, and generally charge fees significantly lower than in other European countries, the most relevant expense for university students are living and travel costs. For this reason, the proximity of a law school may influence the choice of the university subject by Italian students.

Formally, our instrument is:

$$IV_{it} = \frac{1}{\frac{1}{3\sum_{j} d_{ijt}}} \tag{4}$$

where  $d_{ijt}$  are the three distances with the closest law courses that allow to undertake the exam to become lawyer. As discussed in Section 2, access to the legal profession in Italy requires a five-year first degree in law, followed by a two-year apprenticeship in a lawyers office and a successful exam for professional qualification, held once a year in each district of justice. For this reason, we use the lagged proximity. In particular, we consider the localization of law courses and law schools eight year before any corresponding data for the number of lawyers and litigation rates. It is important to stress that our instrument is time variant for two reasons: a) during the period considered new law schools or faculties have been set up and b) new faculties have started offering courses in law.

#### 5.2 Results

Once equipped with these instruments for lawyers, we proceed to analyze the effects on civil litigation rates. Table 4 shows the results of our IV estimation that include province- and year-fixed effect. The first stage regression confirms that our instruments fit well. Both predicted lawyers and proximity are strongly significant and with the expected sign. IV

 $<sup>^{22}</sup>$ Makovec (2005) and Brunello and Cappellari (2008) document that three quarters of the university students in Italy attend a university degree in the same city where their parents live.

diagnostic shows the relevance of the instruments. The F-statistic of the regression is equal to 379.86, which is well above the lower bounds indicated by the literature on weak instruments (see Bound and Holzer (2000) and Stock and Yogo (2002)). Moreover, the validity of our instruments is confirmed by the Hansen-J test for over-identifying restrictions which fails to reject the null hypothesis that the instrument are valid in all cases.

Overall, 2SLS estimates are qualitatively and quantitatively consistent with the OLS results. In particular, according to our IV estimates a 10% increase in the number of lawyers over population is associated with a 3% increase of total litigation rate, a 1.6% increase of litigation rate in civil courts of justice, a 4.5% increase of litigation rate in front of honorary judges of peace and 6% increase in litigation related to compensation.

As already mentioned, in order to test the robustness of our 2SLS estimates, we have performed our IV regressions using as initial number of lawyers the distribution of lawyers across provinces in subsequent years (1993, 1994 and 1995) and, instead of using the average proximity from the three closest faculties offer a law degree, we have considered the average proximity from the three closest Faculties of Law. Results (available on request from the authors) are very robust to different specifications and the magnitude and the significance of our coefficients have not been affected.

Overall, our results suggest that there is a causal effect of the number of lawyers on the civil litigation rates. Therefore our estimates allow us to exclude the reverse causality explanation and to support the SID hypothesis.

# 6 Robustness checks

In this section, we perform several alternative specifications designed to test the robustness of our estimates.

A first possible objection may be related to the fact that our estimates may be picking some spurious effects that are not attributable to the lawyers themselves, but to a systematic change in the economic structure of professions in Italy. In particular, as already discussed in the introduction, also professional accountants and business consultants have experienced a significant increase over the last 15 years. Thus, in order to support or reject such an objection, we re-run our regressions by using the number of professional accountants and business consultants, rather than lawyers. The obtained results from such a regression, presented in table 5, show that the effect of the number of professional accountants and business consultants on civil litigation is not statistically different from zero, thus providing no ground for the above objection.

Another possible objection could be related to the specific type of variables for the litigation rates used in our estimates. In particular, our empirical analysis focuses on litigation as reflected in civil courts, and therefore completely disregard the number of proceedings pending in front of penal courts. Concerning this point, there is a crucial difference between civil and penal cases in Italy. In fact, while a civil case may be started by any citizen, assisted by a lawyer, and is therefore a natural candidate to test the SID hypothesis, a penal file have to be started by a public prosecutor, thus leaving no room for inducement by lawyers. Therefore, if we consider some measures of litigation in penal courts, we expect to see no effects exerted by the number of lawyers. In our robustness check, we consider, for instance, three type of very common property crimes: thefts, car thefts, and bag snatches. Our estimates, presented in table 6, show that the number of lawyers in the court indeed does not exert any significant effect on the number of penal files, thus giving further support to our results.

Finally, we add to our main specification the lagged length of first-instance trials as a control, in order to test whether the expected length of the trial may affect the individual decision to access to civil court. Our estimates, presented in table 7, confirm all main results discussed above and show that the length of first-instance trials is negatively correlated to litigation rates, although with a small effect.

# 7 Conclusion

We have explored whether access to courts in Italy may be driven by the relative number of lawyers active in the provincial courts of justice, in a typical supplier-induced demand frame. We have collected data from different official sources and built an original dataset on the 169 Italian courts of justice, considered in the period 2000-2007. Using panel data estimation techniques, we have investigated the relationship between the local litigation rate and access to the Italian courts of justice, and the number of lawyers officially active in the courts. We have controlled, among others, for the number of magistrates in the court, major economic and socio-demographic characteristics of population, levels of education and social capital in the province. We have addressed the endogeneity issue by constructing two original instrumental variables: the first, based on the fixed coefficient approach, uses national levels of lawyers and lagged values of the local distribution of lawyers across provinces; the second uses the average proximity of the province to the three closest law schools.

Our main result is that the number of lawyers operating in a court does exert a positive and statistically significant effect on the litigation rate. A 10% increase in the number of lawyers over population is associated with a 1.6% to 6% in litigation rates. This effect is robust across several specifications and checks, both on the control variables and the instruments. Our results thus support the SID hypothesis for the Italian lawyers. In particular, our evidence supports the idea that, following an increase in their relative number, lawyers may exploit their informational advantage to induce clients to access to courts even when litigation is unnecessary or ineffective. Our results thus suggest that a share of the massive inflow of new cases started in front of Italian civil courts of justice might be induced by lawyers under an increased competitive pressure. It would be interesting to check whether the number of lawyers operating in a court of justice could have also affected the length of the civil proceedings. This is indeed the aim of our current research.

# References

- Adam, H. (1983). Ambulatory Medical Services and Physicians' Density: Supplier-Induced Demand in Ambulatory Medical Care. Berlin: Dunker and Humblot.
- Autorità Garante della Concorrenza e del Mercato (1997). Relazione Conclusiva dell'Indagine Conoscitiva sul Settore degli Ordini e Collegi Professionali. Technical report.
- Autorità Garante della Concorrenza e del Mercato (2009). Relazione Conclusiva dell'Indagine Conoscitiva sul Settore degli Ordini e Collegi Professionali. Technical report.
- Bartik, T. (1991). Who Benefits from State and Local Economic Development Policies? Mimeo, Kalamazoo, MI: W.E. Upjohn Institute for Employment Research.
- Biancofiore, G. (2009). Al 31 Dicembre 2008 gli Avvocati Erano 198.041: Non Sembrano Pochi. Evoluzione del Numero degli Iscritti. La Previdenza Forense 2, 153–166.
- Blanchard, O. J. and L. F. Katz (1992). Regional Evolutions. Brookings Papers on Economic Activity 23(1), 1–76.
- Bound, J. and H. Holzer (2000). Demand shifts, population adjustment, and labor market outcomes. *Journal of Labor Economics* 1, 20–54.
- Breyer, F. (1984). Supplier-Induced Demand for Medical Services and the Target-Income Hypothesis. Jahrbucher fur Nationalokonomie und Statistik 199, 415–432.
- Breyer, F., H. Muhkenkamp, and H. Adam (1986). Determinants of the Utilization of Physician Services in the System of Statutory Health Insurance in Germany. In J. Schulenburg and v.d. Graf (Eds.), *Essay in Social Security Economics*, pp. 196–216. Berlin: Springer.
- Brunello, G. and L. Cappellari (2008). The labour market effects of Alma Mater: Evidence from Italy. *Economics of Education Review* 27(5), 564–574.
- Buonanno, P., D. Montolio, and P. Vanin (2009). Does Social Capital Reduce Crime? Journal of Law & Economics 52(1), 145–170.
- Card, D. (1993). Using Geographic Variation in College Proximity to Estimate the Return to Schooling. NBER Working Papers 4483.
- Card, D. (2001). Immigrant Inflows, Native Outflows, and the Local Labor Market Impacts of Higher Immigration. *Journal of Labor Economics* 19(1), 22–64.
- CEPEJ (2008). European Judicial System: Efficiency and Quality of Justice. Technical report, Strasbourg.

- Clemenz, G. and K. Gugler (2000). Macroeconomic Development and Civil Litigation. *European Journal of Law and Economics* 9(3), 215–230.
- Cockx, B. and C. Brasseur (2003). The demand for physician services: Evidence from a natural experiment. *Journal of Health Economics* 22(6), 881–913.
- Cortes, P. (2008). The Effect of Low-Skilled Immigration on U.S. Prices: Evidence from CPI Data. *Journal of Political Economy* 116(3), 381–422.
- Coviello, D., A. Ichino, and N. Persico (2009). Giudici in affanno. Mimeo, University of Bologna.
- Darby, M. R. and E. Karni (1973). Free Competition and the Optimal Amount of Fraud. Journal of Law & Economics 16(1), 67–88.
- Donella, D. (2009). Avvocati Clandestini per la Cassa. La Previdenza Forense 2, 168–171.
- Evans, R. (1974). Supplier-Induced Demand: Some Empirical Evidence and Implications. In M. Perlman (Ed.), *The Economics of Health and Medical Care*, pp. 162–173. New York: Macmillan.
- Freeman, R. B. (1980). An Empirical Analysis of the Fixed Coefficient 'Manpower Requirements' Model, 1960-1970. Journal of Human Resources 15(2), 176–199.
- Fuchs, V. R. (1978). The Supply of Surgeons and the Demand for Operations. Journal of Human resources 13(1), 35–56.
- Galizzi, M. M. and M. Miraldo (2009). Reference Prices, Pricing Strategies and Collusion: An Experimental Analysis. Mimeo, Imperial College Business School.
- Ginsburg, T. and G. Hoetker (2006). The Unreluctant Litigant? An Empirical Analysis of Japan's Turn to Litigation. *Journal of Legal Studies* 35, 31–59.
- Gould, E. D., B. A. Weinberg, and D. B. Mustard (2002). Crime Rates And Local Labor Market Opportunities In The United States: 1979-1997. The Review of Economics and Statistics 84(1), 45–61.
- Gruber, J. and M. Owings (1996). Physician Financial Incentives and Cesarean Section Delivery. RAND Journal of Economics 27(1), 99–123.
- Grytten, J., F. Carlsen, and R. Sorensen (1995, June). Supplier inducement in a public health care system. *Journal of Health Economics* 14(2), 207–229.
- Grytten, J., D. Holst, and P. Laake (1990). Supplier inducement : Its effect on dental services in Norway. *Journal of Health Economics* 9(4), 483–491.

- Hanssen, F. A. (1999). The Effect of Judicial Institutions on Uncertainty and the Rate of Litigation: The Election versus Appointment of State Judges. *Journal of Legal Studies* 28(1), 205–32.
- IMF (2009). World Economic Outlook Database. Technical report, Washington.
- Kraft, K., J. Schulenber, and v. Graf (1986). Co-Insurance and Supplier-Induced Demand in Medical Care: What Do We Have to Expect as the Physicians Response to Increased Out-of-Pocket Payments? Zeitschrift fur die gesamte Staatwissenschaft 142, 360–379.
- Madden, D., A. Nolan, and B. Nolan (2005). GP reimbursement and visiting behaviour in Ireland. *Health Economics* 14(10), 1047–1060.
- Makovec, M. (2005). Does it pay to study far from home? Explaining the returns to geographic mobility of Italian college graduates. Mimeo, Bocconi University.
- Marchesi, D. (2003). Litiganti, Avvocati e Magistrati. Bologna.
- Miraldo, M. (2009). Reference pricing and firms' pricing strategies. *Journal of Health Economics* 28(1), 176–197.
- Posner, R. (1997). Explaining the Variance in the Number of Tort Suits across U.S. States and between the United States and England. *Journal of Legal Studies* 26, 477–489–600.
- Rossiter, L. and G. Wilensky (1984). Identification of physician-induced demand. *Journal of Human resources 19*, 231–244.
- Sobbrio, G., E. D'Agostino, and E. Sironi (2009). New Disputes and Delay in Italian Courts. Siep working paper.
- Stock, J. H. and M. Yogo (2002, November). Testing for Weak Instruments in Linear IV Regression. NBER Technical Working Papers 0284, National Bureau of Economic Research, Inc.
- Tussing, A. (1983). Physicians-induce Demand for Medical Care: Irish General Practitioners. The Economic and Social Review 14, 225–247.
- Van De Voorde, C., E. Van Doorslaer, and E. Schokkaert (2001). Effects of cost sharing on physician utilization under favourable conditions for supplier-induced demand. *Health Economics* 10(5), 457–471.

Table 1:	Descriptive	statistics
----------	-------------	------------

	obs.	mean	std. dev.	min	max
Total litigation	824	2.418	.450	1.464	4.200
Civil courts	824	1.816	.306	.955	2.734
Judges of peace	824	1.510	.705	149	4.078
Compensation	824	.597	.889	-1.215	3.664
Lawyers	824	1.656	.697	.126	6.995
Judges	824	.075	.0264	.024	.178
Accountants	824	.616	.229	.005	1.874
Density	824	246.027	331.301	22.954	$2,\!640.92$
Concentration	824	47.376	74.562	9.593	680.92
GDP	824	$18,\!121.36$	$4,\!559.8$	9,829.364	30,370.54
Employment	824	44.751	6.774	28.184	58.662
High school	824	.338	.039	.257	.485
Associations	824	33.727	16.340	7.75	104.73

*Note:* This table reports the descriptive statistics for all dependent and explanatory variables across the 103 Italian provinces during the period 2000-2007.

	Total litigation	Civil courts	Judges of peace	Compensation	Lawyers	Judges	Density	Concentration	GDP	High school	Associations
Total litigation	1.0000										
$Civil \ courts$	0.6211	1.0000									
Judges of peace	0.9447	0.3798	1.0000								
Lawyers	0.5072	0.4779	0.5197	1.0000							
Judges	0.5951	0.3701	0.5987	0.6306	0.4018	1.0000					
Density	0.2808	0.1581	0.2573	0.3106	0.2180	0.2609	1.0000				
Concentration	0.1378	0.2994	0.0650	0.1002	0.1380	0.2096	0.3458	1.0000			
GDP	-0.4869	-0.1165	-0.5449	-0.5787	-0.1669	-0.5022	0.1856	0.1891	1.0000		
High school	0.0779	0.1730	0.0342	0.0324	0.3558	0.0239	0.0231	0.0745	0.0188	1.0000	
Associations	-0.2140	0.0659	-0.2611	-0.2393	-0.1379	-0.1776	0.0179	0.3954	0.4713	-0.1482	1.0000

matrix
Correlation
;;
Table

	Total litigation rate	Civil courts	Judges of peace	Compensation	Total litigation rate	Civil courts	Judges of peace	Compensation
	(1)	(2)	(3)	(4)	(5)	(9)	(2)	(8)
Lawyers	.2917*** (.0866)	$.1669^{**}$ (.0812)	$.4032^{***}$ (.1029)	$.5097^{***}$ (.1229)	$.2880^{***}$ (.0905)	$.1616^{**}$ (.0818)	$.3907^{***}$ (.1121)	.4887*** (.1318)
Judges					2789 (.9592)	2233 (.7016)	0803 (1.2171)	.8175 (.7949)
High School					$-1.9723^{*}$ (1.0468)	$-2.6510^{***}$ (.9250)	-1.9288 (1.4594)	7606 (1.3635)
Associations					$.0050^{**}$ (.0019)	$.0034^{*}$ (.0018)	.0031 (.0027)	0016 (.0031)
GDP per capita					.00002 (.0002)	7.00e-06 (1.00e-05)	$.00004^{*}$ (.0002)	.00002 (.0002)
Employment					.0080 (.0050)	.0060 (.0042)	.0068	.0066 (.0072)
Density					$0030^{**}$ (.0013)	0011 (.0010)	$0051^{**}$ (.0020)	$0052^{***}$ (.0018)
Concentration					.0128** (.0050)	$.0067^{*}$	$.0206^{***}$ (.0080)	$.0219^{**}$ (.0107)
Obs.	824	824	824	824	824	824	824	824
$\operatorname{Provinces}$	103	103	103	103	103	103	103	103
Prov. FE	Yes	$\mathbf{Y}_{\mathbf{es}}$	Yes	$\mathbf{Yes}$	Yes	$\mathbf{Yes}$	${ m Yes}$	$\mathbf{Y}_{\mathbf{es}}$
Year FE	Yes	Yes	Yes	Yes	Yes	$\mathbf{Yes}$	$\mathbf{Yes}$	Yes
<i>Note:</i> This table dependent variable number of lawyers respectively. All o presented in parer respectively.	presents the results ( e is the log of civil li s over province popul ther variables are del thesis. *, ** and **	of OLS estima tigation rate r lation. The so fined in the A ** denote rejec	tes on a panel of ecorded by the ci urces of data for ppendix. Provinc ttion of the null 1	ty yearly observat vil courts of jus lawyers and civ e and year fixed hypothesis of th	ions for all 103 Italia tice, for each category il litigation are <i>Casse</i> effects are included i e coefficient being eq	n provinces d v of civil litigs <i>uforense</i> (Law. n all specificat ual to 0 at 10	uring the period ation. The variab yer Pension Agen tions. Robust sta %, 5% and 1% si	2000-2007. The le lawyers is the cy) and ISTAT, and arrors are gnificance level,

Table 3: Panel regressions: baseline

30

	Panel A: First-stage	
Predicted lawyers	$\frac{1.0032^{***}}{(.0419)}$	
Proximity	$0.0665^{**}$ (.0318)	

379.86

#### F-test

		Panel B: Second-stage					
	Total litigation rate	Civil courts	Judges of peace	Compensation			
	(1)	(2)	(3)	(4)			
Lawyers	$.3099^{***}$ (.0603)	$.1664^{***}$ (.0593)	$.4539^{***}$ (.0858)	.6086*** (.1061)			
Judges	2883 (.8780)	2253 (.6473)	1075 (1.1779)	.7660 (.7833)			
High school	$-1.9449^{***}$ (.7262)	$-2.6451^{***}$ (.7085)	$-1.8499^{*}$ (.9987)	6108 (.9066)			
Associations	$.0051^{***}$ (.0017)	$.0034^{**}$ (.0016)	.0033 $(.0023)$	0011 (.0024)			
GDP per capita	.00002** (1.00e-05)	7.00e-06 (1.00e-05)	.00004*** (1.00e-05)	.00002 $(1.00e-05)$			
Employment	.0079* (.0041)	.0060 (.0039)	.0065 $(.0053)$	.0059 $(.0051)$			
Density	0029*** (.0007)	0011 (.0007)	$0050^{***}$ (.0010)	$0050^{***}$ (.0011)			
Concentration	.0129*** (.0034)	.0067** (.0032)	.0209*** (.0043)	$.0225^{***}$ (.0056)			
Hansen-J test	0.037	1.144	0.066	0.660			
Obs.	824	824	824	824			
Provinces	103	103	103	103			
Prov. FE	Yes	Yes	Yes	Yes			
Year FE	Yes	Yes	Yes	Yes			

*Note:* The top panel of this table presents first-stage estimates of IV regressions. The bottom panel reports the results of 2SLS (second-stage) estimates on on a panel of yearly observations for all 103 Italian provinces during the period 2000-2007. The dependent variable is the log of civil litigation rate recorded by the civil courts of justice, for each category of civil litigation. All control variables in Table 3 are always included, both in the first and second stage. The sources of data for lawyers and civil litigation are *Cassaforense* (Lawyer Pension Agency) and ISTAT, respectively. All other variables are defined in the Appendix. The F-statistic refers to the null hypothesis that the coefficient on the excluded instruments are jointly equal to zero in the first stage. The Hansen J-test is a test of overidentifying restrictions, distributed as chi-square under the null of instrument validity. Robust standard errors are presented in parenthesis. \*, \*\* and \*\*\* denote rejection of the null hypothesis of the coefficient being equal to 0 at 10%, 5% and 1% significance level, respectively.

	Total litigation rate	Civil courts	Judges of peace	Compensation
	(1)	(2)	(3)	(4)
Accountants	.1807	0512	.3843	.2770
	(.2130)	(.2553)	(.2730)	(.3761)
Judges	2142	0860	1013	.8997
	(.9958)	(.7016)	(1.2702)	(.8982)
High School	$-2.2325^{**}$	$-2.7624^{***}$	$-2.3822^{*}$	-1.3471
	(1.0226)	(.8898)	(1.4345)	(1.3787)
Associations	.0041**	.0026	.0021	0031
	(.0020)	(.0018)	(.0028)	(.0036)
GDP per capita	.00002	9.00e-06	$.00004^{*}$	.00002
	(.00002)	(1.00e-05)	(.00002)	(.00002)
Employment	$.0091^{*}$	.0065	.0084	.0086
	(.0053)	(.0044)	(.0075)	(.0079)
Density	0034*	0014	0057**	0058***
·	(.0018)	(.0014)	(.0026)	(.0022)
Concentration	$.0123^{*}$	.0066	$.0194^{*}$	.0203
	(.0069)	(.0049)	(.0104)	(.0141)
Obs.	824	824	824	824
Provinces	103	103	103	103
Prov. FE	Yes	Yes	Yes	Yes
Year FE	Yes	Yes	Yes	Yes

Table 5: Robustness check: Professional accountants

*Note:* This table presents the results of OLS estimates on a panel of yearly observations for all 103 Italian provinces during the period 2000-2007. The dependent variable is the log of civil litigation rate recorded by the civil courts of justice, for each category of civil litigation. The variable accountants is the number of professional accountants over province population. The sources of data for accountants and civil litigation are CNPADC (Professional Accountants Pension Agency) and ISTAT, respectively. All other variables are defined in the Appendix. Province and year fixed-effects are included in all specifications. Robust standard errors are presented in parenthesis. \*, \*\* and \*\*\* denote rejection of the null hypothesis of the coefficient being equal to 0 at 10%, 5% and 1% significance level, respectively.

	Theft	Car theft	Bag snatch	Murder
	(1)	(2)	(3)	(4)
Lawyers	.0654	.1092	.0619	0701
	(.0410)	(.0936)	(.1156)	(.1561)
Judges	.1488	$1.8552^{**}$	.3145	.3577
	(.4863)	(.8819)	(1.0840)	(2.1031)
High School	$1.5625^{*}$	.1736	2640	$4.1036^{*}$
	(.9415)	(2.1689)	(2.0220)	(2.4933)
Associations	.0002	.0007	0015	0018
	(.0019)	(.0038)	(.0042)	(.0067)
GDP per capita	8.81e-08	.00003	-6.00e-06	-1.00e-05
	(1.00e-05)	(.00002)	(.00002)	(.00004)
Employment	.0030	.0055	.0045	.0072
	(.0035)	(.0110)	(.0106)	(.0162)
Density	$.0015^{*}$	.0023	0073**	.0023
	(.0008)	(.0014)	(.0032)	(.0020)
Concentration	$.0061^{*}$	$.0105^{*}$	$.0204^{***}$	.0067
	(.0034)	(.0057)	(.0055)	(.0092)
Obs.	824	824	824	824
Provinces	103	103	103	103
Prov. FE	Yes	Yes	Yes	Yes
Year FE	Yes	Yes	Yes	Yes

Table 6: Robustness check: Crime rates

Note: This table presents the results of OLS estimates on a panel of yearly observations for all 103 Italian provinces during the period 2000-2007. The dependent variable is the log of crimes reported by the police over the total population, for each category of criminal offense. The variable lawyers is the number of lawyers over province population. The sources of data for lawyers and crime rates are *Cassaforense* (Lawyer Pension Agency) and ISTAT, respectively. All other variables are defined in the Appendix. Province and year fixed-effects are included in all specifications. Robust standard errors are presented in parenthesis. \*, \*\* and \*\*\* denote rejection of the null hypothesis of the coefficient being equal to 0 at 10%, 5% and 1% significance level, respectively.

	Total litigation rate	Civil courts	Judges of peace	Compensation
	(1)	(2)	(3)	(4)
Lawyers	.2659*** (.0847)	.2131*** (.0776)	.3201*** (.1094)	.4610*** (.1382)
Judges	.2893 (.9360)	.1015 (.6202)	.6242 (1.2203)	$2.0735^{**}$ (.9764)
$\mathrm{Length}_{-1}$	$0001^{***}$ (.00004)	0001*** (.00005)	0001* (.00007)	00008 (.00009)
High school	-1.1699 (.9825)	$-2.0785^{**}$ (.8233)	-1.1614 (1.4890)	9974 (1.3574)
Associations	$.0054^{***}$ (.0020)	$.0030^{*}$ (.0016)	.0046(.0029)	0006 (.0031)
GDP per capita	.00004** (.00002)	.00002 (.00002)	.00005* (.00002)	.00003 (.00003)
Employment	.0064 (.0051)	.0045 (.0043)	.0055 (.0076)	.0060 (.0074)
Density	0027** (.0013)	0014 (.0011)	0044** (.0020)	$0051^{***}$ (.0018)
Concentration	.0093** (.0043)	.0053* (.0028)	$.0163^{**}$ (.0079)	.0230** (.0098)
Obs.	721	721	721	721
Provinces	103	103	103	103
Prov. FE	Yes	Yes	Yes	Yes
Year FE	Yes	Yes	Yes	Yes

Table 7: Robustness check: First-instance proceedings lenght

*Note:* This table presents the results of OLS estimates on a panel of yearly observations for all 103 Italian provinces during the period 2001-2007. The dependent variable is the log of crimes reported by the police over the total population, for each category of criminal offense. The variable lawyers is the number of lawyers over province population. The sources of data for lawyers and crime rates are *Cassaforense* (Lawyer Pension Agency) and ISTAT, respectively. All other variables are defined in the Appendix. Province and year fixed-effects are included in all specifications. Robust standard errors are presented in parenthesis. \*, \*\* and \*\*\* denote rejection of the null hypothesis of the coefficient being equal to 0 at 10%, 5% and 1% significance level, respectively.

# **Discussion Papers recentemente pubblicati**

# Anno 2006

0601 – Francesco **MENONCIN** "The role of longevity bonds in optimal portfolios" (gennaio)

0602 – Carmine **TRECROCI**, Matilde **VASSALLI** "Monetary Policy Regime Shifts: New Evidence from Time-Varying Interest-Rate Rules" (gennaio)

0603 – Roberto CASARIN, Carmine TRECROCI "Business Cycle and Stock Market Volatility: A Particle Filter Approach" (febbraio)

0604 – Chiara DALLE NOGARE, Matilde VASSALLI "A Pressure-Augmented Taylor Rule for Italy" (marzo)

0605 – Alessandro **BUCCIOL**, Raffaele **MINIACI** "Optimal Asset Allocation Based on Utility Maximization in the Presence of Market Frictions" (marzo)

0606 – Paolo M. **PANTEGHINI** "The Capital Structure of Multinational Companies under Tax Competition" (marzo)

0607 – Enrico MINELLI, Salvatore MODICA "Credit Market Failures and Policy" (gennaio)

0608 – J.H. DRÈZE, E. MINELLI, M. TIRELLI "Production and Financial Policies Under Asymmetric Information" (febbraio)

0609 – Françoise FORGES, Enrico MINELLI "Afriat's Theorem for General Budget Sets" (marzo)

0610 – Aviad HEIFETZ, Enrico MINELLI "Aspiration Traps" (marzo)

0611 – Michele **MORETTO**, Paolo M. **PANTEGHINI**, Carlo **SCARPA** "*Profit Sharing and Investment by Regulated Utilities: a Welfare Analysis*" (aprile)

0612 – Giulio **PALERMO** "Il potere come relazione sociale. Il caso dell'università baronale italiana" (giugno)

0613 – Sergio VERGALLI "Dynamics in Immigration Community" (luglio)

0614 – Franco SPINELLI, Carmine TRECROCI "Maastricht: New and Old Rules" (luglio)

0615 – Giulio **PALERMO** "La valutazione dei titoli scientifici dei docenti del Dipartimento di Scienze Economiche dell'Università di Brescia" (settembre)

0616 – Rosella **LEVAGGI** "*Tax evasion and the cost of public sector activities*" (settembre)

0617 – Federico **BOFFA**, Carlo **SCARPA** "*Exporting Collusion under Capacity Constraints: an Anti-Competitive Effect of Market Integration*" (ottobre)

0618 – Monica **BILLIO**, Roberto **CASARIN** "Stochastic Optimisation for Allocation Problems with Shortfall Risk Constraints" (ottobre)

# Anno 2007

0701 - Sergio VERGALLI "Entry and Exit Strategies in Migration Dynamics" (gennaio)

0702 – Rosella **LEVAGGI**, Francesco **MENONCIN** "A note on optimal tax evasion in the presence of merit goods" (marzo)

0703 – Roberto CASARIN, Jean-Michel MARIN "Online data processing: comparison of Bayesian regularized particle filters" (aprile)

0704 – Gianni AMISANO, Oreste TRISTANI "Euro area inflation persistence in an estimated nonlinear DSGE model" (maggio)

0705 – John **GEWEKE**, Gianni **AMISANO** *"Hierarchical Markov Normal Mixture Models with Applications to Financial Asset Returns"* (luglio)

0706 – Gianni AMISANO, Roberto SAVONA "Imperfect Predictability and Mutual Fund Dynamics: How Managers Use Predictors in Changing Systematic Risk" (settembre)

0707 – Laura **LEVAGGI**, Rosella **LEVAGGI** "*Regulation strategies for public service provision*" (ottobre)

# Anno 2008

- 0801 Amedeo **FOSSATI**, Rosella **LEVAGGI** "Delay is not the answer: waiting time in health care & income redistribution" (gennaio)
- 0802 Mauro **GHINAMO**, Paolo **PANTEGHINI**, Federico **REVELLI** "*FDI determination and corporate tax competition in a volatile world*" (aprile)
- 0803 Vesa **KANNIAINEN**, Paolo **PANTEGHINI** "Tax neutrality: Illusion or reality? The case of Entrepreneurship" (maggio)
- 0804 Paolo **PANTEGHINI** "Corporate Debt, Hybrid Securities and the Effective Tax Rate" (luglio)
- 0805 Michele **MORETTO**, Sergio **VERGALLI** "Managing Migration Through Quotas: an Option-Theory perspective" (luglio)
- 0806 Francesco **MENONCIN**, Paolo **PANTEGHINI** "The Johansson-Samuelson Theorem in General Equilibrium: A Rebuttal" (luglio)
- 0807 Raffaele **MINIACI** Sergio **PASTORELLO** "Mean-variance econometric analysis of household portfolios" (luglio)
- 0808 Alessandro **BUCCIOL** Raffaele **MINIACI** "Household portfolios and implicit risk aversion" (luglio)
- 0809 Laura **PODDI**, Sergio **VERGALLI** "Does corporate social responsability affect firms performance?" (luglio)
- 0810 Stefano CAPRI, Rosella LEVAGGI "Drug pricing and risk sarin agreements" (luglio)
- 0811 Ola ANDERSSON, Matteo M. GALIZZI, Tim HOPPE, Sebastian KRANZ, Karen VAN DER WIEL, Erik WENGSTROM "Persuasion in Experimental Ultimatum Games" (luglio)
- 0812 Rosella **LEVAGGI** "Decentralisation vs fiscal federalism in the presence of impure public goods" (agosto)
- 0813 Federico **BIAGI**, Maria Laura **PARISI**, Lucia **VERGANO** "Organizational Innovations and Labor Productivity in a Panel of Italian Manufacturing Firms" (agosto)
- 0814 Gianni AMISANO, Roberto CASARIN "Particle Filters for Markov-Switching Stochastic-Correlation Models" (agosto)
- 0815 Monica **BILLIO**, Roberto **CASARIN** "Identifying Business Cycle Turning Points with Sequential Monte Carlo Methods" (agosto)
- 0816 Roberto CASARIN, Domenico SARTORE "Matrix-State Particle Filter for Wishart Stochastic Volatility Processes" (agosto)
- 0817 Roberto CASARIN, Loriana PELIZZON, Andrea PIVA "Italian Equity Funds: Efficiency and Performance Persistence" (settembre)
- 0818 Chiara **DALLE NOGARE**, Matteo **GALIZZI** "*The political economy of cultural spending: evidence from italian cities*" (ottobre)

# Anno 2009

0901 – Alessandra **DEL BOCA**, Michele **FRATIANNI**, Franco **SPINELLI**, Carmine **TRECROCI** "*Wage Bargaining Coordination and the Phillips curve in Italy*" (gennaio)

- 0902 Laura **LEVAGGI**, Rosella **LEVAGGI** "Welfare properties of restrictions to health care services based on cost effectiveness" (marzo)
- 0903 Rosella **LEVAGGI** "From Local to global public goods: how should externalities be represented?" (marzo)
- 0904 Paolo **PANTEGHINI** "On the Equivalence between Labor and Consumption Taxation" (aprile)
- 0905 Sandye **GLORIA-PALERMO** "Les conséquences idéologiques de la crise des subprimes" (aprile)
- 0906 Matteo M. GALIZZI "Bargaining and Networks in a Gas Bilateral Oligopoly" (aprile)

0907 – Chiara **D'ALPAOS**, Michele **FRATIANNI**, Paola **VALBONESI**, Sergio **VERGALLI** "*It is never too late*": *optimal penalty for investment delay in public procurement contracts* (maggio)

0908 – Alessandra **DEL BOCA**, Michele **FRATIANNI**, Franco **SPINELLI**, Carmine **TRECROCI** "*The Phillips Curve and the Italian Lira*, 1861-1998" (giugno)

0909 – Rosella **LEVAGGI**, Francesco **MENONCIN** "Decentralised provision of merit and impure public goods" (luglio)

0910 – Francesco **MENONCIN**, Paolo **PANTEGHINI** "*Retrospective Capital Gains Taxation in the Real World*" (settembre)

0911 – Alessandro **FEDELE**, Raffaele **MINIACI** "Do Social Enterprises Finance Their Investments Differently from For-profit Firms? The Case of Social Residential Services in Italy" (ottobre)

0912 – Alessandro FEDELE, Paolo PANTEGHINI, Sergio VERGALLI "Optimal Investment and Financial Strategies under Tax Rate Uncertainty" (ottobre)

0913 – Alessandro **FEDELE**, Francesco **LIUCCI**, Andrea **MANTOVANI** "Credit Availability in the Crisis: The Case of European Investment Bank Group" (ottobre)